Author's response to reviews

Title: MicroRNA expression as risk biomarker of breast cancer metastasis: a pilot retrospective case-cohort study.

Authors:

Augusto L.F. Marino (augusto.marino@gmail.com)
Adriane F. Evangelista (adriane.feijo@gmail.com)
Rene A.C Vieira (reneacy@terra.com.br)
Taciane Macedo (macedo.taciane@gmail.com)
Ligia M Kerr (kerr.l@hotmail.com)
Lucas FA Machado (lucasfariamachado@yahoo.com)
Adhemar Longatto Filho (longatto16@hotmail.com)
Henrique C.S Silveira (henriquecssilveira@gmail.com)
Márcia M.C Marques (mmcmsilveira@gmail.com)

Version: 5  Date: 24 September 2014

Author's response to reviews: see over
Dear Editor:

Please find enclosed our revised manuscript, “MicroRNA expression as risk biomarker of breast cancer metastasis: a pilot retrospective case-cohort study.” by Marino et al., that was accepted to be published as research article after following revisions.

We included as requested the conclusion section (page 21).

We look forward to hearing from you at your earliest convenience.

Best Regards,

Marcia M.C.Marques Silveira